Author’s response to reviews

Title: Challenges and Opportunities for Effective Adoption of HRH Information Systems in Developing Countries: National rollout of HRHIS and TIIS in Tanzania

Authors:

Hisahiro Ishijima (hisahiro.ishijima@gmail.com)
Martin Mapunda (martinstive@yahoo.com)
Mathew Mndeme (mathewmndeme@gmail.com)
Felix Sukumus (felix@entersoftsystems.com)
Violeth Mlay (lalethmlay25@gmail.com)

Version: 2 Date: 28 October 2014

Author’s response to reviews:

Dear Editor of the Human Resources for Health

We hereby submit the revised manuscript of MS: 1098646867147561, “Challenges and Opportunities for Effective Adoption of HRH Information Systems in Developing Countries: National rollout of HRHIS and TIIS in Tanzania”, based on the instruction from the Journal Editorial Office.

We hope that the revised manuscript covered all requirements for peer review process.

Thank you for your consideration of the manuscript.

Sincerely,

Hisahiro Ishijima